

April 29, 2022

Reviewer 1:

This is a very interesting case. The author describes the diagnosis and treatment process of the patient in detail, which provides a certain reference for the diagnosis of celiac disease.

Reviewer 2:

Dear Editor, Thanks to the authors for this case report. There is limited studies correlating infectious agents such as EBV, VZV to virus. I think that It will contribute to the literature.

Dear Reviewers,

Thank you for the opportunity to revise. The title has been changed, an addition made to the abstract, and figures presented in power point.

Thanks

Lisa Maness

336-253-3677

manesslisa@gmail.com